Author’s response to reviews

Title: Clustering in surgical trials - database of intracluster correlations

Authors:

Jonathan A Cook (j.a.cook@abdn.ac.uk)
Thomas Bruckner (bruckner@imbi.uni-heidelberg.de)
Graeme S MacLennan (g.maclennan@abdn.ac.uk)
Christoph M Seiler (christoph.seiler@med.uni-heidelberg.de)

Version: 5 Date: 4 January 2012

Author’s response to reviews:

The changes made are to correct typological and grammatical errors or to address inconsistencies. No substantive changes have been made.